Author’s response to reviews

Title: Atypical presentation of Renal Cell Carcinoma: A case report

Authors:

Deepak D Doshi Mr (drdeepdoshi@yahoo.com)
Michael M Saab Mr (michael.saab@pat.nhs.uk)

Version: 2 Date: 17 February 2007

Author’s response to reviews:

Dear Editor,

Thank you for your email.

We have carefully read your mail, and done all the revisions as suggested. Please let us know if you need clarification any further.

Restructure the title so it ends ‘:A Case Report.’ - Done

Include Figure Legends for all figures - Done

Include an Introduction section - Done

Take the text out of bold. - Taken out

Include a Competing Interests declaration - Done

Include an Authors' Contributions list - Done

Include an Acknowledgements list. This should contain a statement to confirm that patient consent was received from the patient for the manuscript to be published.- Done

Many Thanks
Deepak Doshi

and

M saab, N Singh